Author’s response to reviews

Title: the effect of tobacco, XPC, XPD and XPG genetic variants in bladder cancer development

Authors:

Kamel ROUISSI (rouissik2000@yahoo.fr)
Islem BEN BAHRIA (islemmm@gmail.com)
Karim BOUGATEF (karimhamhama@yahoo.fr)
Raja MARRAKCHI (raja.triki@laposte.net)
Nejla STAMBOULI (nejla.stambouli@topnet.tn)
Khouloud HAMDI (hamdi.khouloud@yahoo.fr)
Mohamed CHERIF (drcherifmed@yahoo.fr)
Mohamed Riadh BEN SLAMA (riadhbenslama@yahoo.com)
Mohamed SFAXI (momosfa@yahoo.fr)
Fethi BEN OTHMAN (Fethi.benothman@yahoo.fr)
Mohamed CHEBIL (Mohamed.Chebil@yahoo.fr)
Amel BENAMMAR ELGAAIED (a_benammarelgaied@yahoo.fr)
Slah OUERHANI (slah_mekni@yahoo.fr)

Version: 3 Date: 10 March 2011

Author’s response to reviews: see over
Major revisions (we require the author to make these changes)

1- Title page: please include a title page in the manuscript file. This should contain; Title, Author list, Affiliations, E-mail addresses.

   Answer to comments 1: The title page was added in the text

2- The abstract: please ensure that the Abstract is present in the manuscript and matches exactly what is in the submission system

   Answer to comment 2: The abstract is present in the manuscript and matches exactly what is in the submission system

Minor revisions

1- Abstract: please rename the patients and methods section 'methods': We have rename the “patients and methods” section as 'methods'

2- Materials and methods: please rename this section 'Methods': we have rename this section

3- Tables: please note that we are unable to correctly display merged cells where the merged cell crosses rows: please re-layout your table without these merged elements

4- Tables- please remove the visible vertical lines from your tables: we have removed the vertical line from the tables

5- Typography: Please take this opportunity to check your manuscript for any typographical errors and to make any final corrections or revisions. This is the final proofing stage for your manuscript, and you will not be able to make any changes after acceptance: We have checked our manuscript for any typographical errors